Author's response to reviews

Title: Comparison of exercise, dobutamine-atropine and dipyridamole-atropine stress echocardiography in detecting coronary artery disease.

Authors:
Ivana P Nedeljkovic (ivannanedeljkovic@yahoo.com)
Miodrag Ostojic (miodra.ostojic@kcs.sc.yu)
Branko Beleslin (branko.beleslin@kcs.ac.yu)
Ana Djordjevic-Dikic (scali@bitsyu.net)
Jelena Stepanovic (jelena.stepanovic@kcs.ac.yu)
Milan Nedeljkovic (milanned@hotmail.com)
Sinisa Stojkovic (sinsisa.stojkovic@kcs.ac.yu)
Goran Stankovic (gorastan@hotmail.com)
Jovica Saponjski (jovica.saponjski@kcs.ac.yu)
Zorica Petrasinovic (zorica.petrasinovic@kcs.ac.yu)
Vojislav Giga (giga@eunet.yu)
Predrag Mitovic (predragm@hotmail.com)

Version: 2 Date: 21 April 2006

Author's response to reviews:

Dear Sires, I kindly ask you to receive this I hope final version of my manuscript. I made all necessary changes on title page (about corresponding author) and abstract, as well as in manuscript text. I also added all necessary information about specific contribution of each author and their particular conflict of interest. I want to thank you for all comments and useful suggestions. The cooperation with Editorial board of CU represents a very good experience for me.
Sincerely

dr Ivana Nedeljkovic